Author’s response to reviews

Title: Sociodemographic correlates of urine culture test utilization in Calgary, Alberta

Authors:

Thomas Griener (thomas.griener@cls.ab.ca)
Christopher Naugler (christopher.naugler@cls.ab.ca)
Wilson Chan (wilson.chan@cls.ab.ca)
Deirdre Church (deirdre.church@albertahealthservices.ca)

Version: 2 Date: 22 Dec 2017

Author’s response to reviews:

The following editorial revisions were suggested and have been made:

1) Abstract:

Please change the ‘Objectives’ header to ‘Background’ to conform to journal style.

Change made as requested. Page 2 line 3.

2) List of abbreviations:

Please move the list to immediately after the main Conclusions section.

Change made as requested. List of abbreviations now Page 10 line 24 to page 11 line 2.

3) Declarations section:

* Ethics approval and consent to participate:

Please change the ‘Ethics’ header to ‘Ethics approval and consent to participate’.

Change made as requested. Page 11 line 7.

* Consent for publication:

Please include this declaration and state: ‘Not applicable’.
* Competing interests:

Please change the ‘Disclosures/conflicts of interest’ header to ‘Competing interests’.

Change made as requested. Page 11 line 21

* Funding:

Please include this declaration with a statement as to how the study was funded, indicating the role of the funding body in: the design of the study; the collection, analysis, and interpretation of data; and the writing of the manuscript.

Declaration now included. Page 11 line 25-27

* Authors’ contributions:

Please add a sentence confirming all authors have read and approve the final manuscript.

Change made as requested. Sentence confirming all authors have read and approve of the final manuscript included Page 12 line 4

* Acknowledgements:

Please move this statement to the Declarations section.

Change made as requested. Acknowledgments now Page 12 lines 6-10

4) Copy-edit:

Please ensure the manuscript is proofread to amend any spelling errors. Please correct the spelling of ‘Yearly’ in the axis label for Figure 1.

The spelling error has been corrected and the document proofread.